Author’s response to reviews

Title: Association of major dietary patterns and different metabolic phenotypes: A population-based study of northwestern Iran

Authors:

Leila Nikniaz (nikniazleila@gmail.com)
Mahdieh Abbasalizad Farhangi (abbasalizadm@tbzmed.ac.ir)
Jafar sadeh Tabrizi (tabrizij@tbzmed.ac.ir)
Zeinab Nikniaz (znikniaz@hotmail.com)

Version: 3 Date: 23 Oct 2019

Author’s response to reviews:

Dear Editor;

All mentioned sections are edited and the abbreviation section is added after conclusion. Moreover, the reviewer asked for gender specific dietary pattern. However, in the present study, our aim was not determination of the gender specific dietary pattern. Instead, we aimed to determine the overall dietary pattern and its correlation with metabolic phenotype. In tables 1 and 2, we did separate analysis for men and women for more clarification and in table 3, according to reviewer suggestion, we adjusted the results for sex in multinomial regression analysis. It would be appreciated if you accept the results and discussion in its current form.